Author's response to reviews

Title: Modified technique for transscleral fixation of posterior chamber intraocular lenses

Authors:

Chongde Long (zoclong@21cn.com)
Yantao Wei (weiyantao75@126.com)
Zhaohui Yuan (jorphin@hotmail.com)
Zhiqing Zhang (zhang-zhiqing@sohu.com)
Xiaofeng Lin (linxiaof@mail.sysu.edu.cn)
Bingqian Liu (bingqianliu@gmail.com)

Version: 5 Date: 18 September 2015

Author's response to reviews:

1. Include all authors' e-mail address on the title page
   Added.

2. Clarify if consent was written or oral.
   Consent was written, already clarified.

3. Remove tables from the additional files and include them at the end of the main ms
   Done.